

# Latitude, Elevation, and Mean Annual Temperature Predict Peat Organic Matter Chemistry at a Global Scale

Brittany A Verbeke, Louis J Lamit, Erik A Lilleskov, Suzanne B Hodgkins, Nathan Basiliko, Evan S Kane, Roxane Andersen, Rebekka R E Artz, Juan C Benavides, Brian W Benscoter, et al.

# ▶ To cite this version:

Brittany A Verbeke, Louis J Lamit, Erik A Lilleskov, Suzanne B Hodgkins, Nathan Basiliko, et al.. Latitude, Elevation, and Mean Annual Temperature Predict Peat Organic Matter Chemistry at a Global Scale. Global Biogeochemical Cycles, 2022, 36 (2), 10.1029/2021gb007057. hal-04075707

# HAL Id: hal-04075707 https://hal.science/hal-04075707

Submitted on 20 Apr 2023  $\,$ 

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



# Global Biogeochemical Cycles<sup>®</sup>

## **RESEARCH ARTICLE**

10.1029/2021GB007057

Brittany A. Verbeke and Louis J. Lamit contributed equally to the work.

#### **Key Points:**

- Peatland soil organic matter stability can be evaluated through the relative abundances of carbohydrate and aromatic content in peat
- Peat at higher latitudes and elevations has larger carbohydrate and smaller aromatic content relative lower latitudes and elevations
- Our results foreshadow a transition of organic matter from higher latitude peatlands to a more recalcitrant form in response to warming

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

J. P. Chanton, jchanton@fsu.edu

#### Citation:

Verbeke, B. A., Lamit, L. J., Lilleskov, E. A., Hodgkins, S. B., Basiliko, N., Kane, E. S., et al. (2022). Latitude, elevation, and mean annual temperature predict peat organic matter chemistry at a global scale. *Global Biogeochemical Cycles*, *36*, e2021GB007057. https://doi. org/10.1029/2021GB007057

Received 26 APR 2021 Accepted 24 JAN 2022

#### **Author Contributions:**

Conceptualization: Brittany A. Verbeke, Louis J. Lamit, Erik A. Lilleskov, Suzanne B. Hodgkins, Jeffrey P. Chanton Data curation: Brittany A. Verbeke, Louis J. Lamit Funding acquisition: Erik A. Lilleskov, Suzanne B. Hodgkins, Jeffrey P. Chanton

© 2022. American Geophysical Union. All Rights Reserved.

# Latitude, Elevation, and Mean Annual Temperature Predict Peat Organic Matter Chemistry at a Global Scale

Brittany A. Verbeke<sup>1</sup>, Louis J. Lamit<sup>2</sup>, Erik A. Lilleskov<sup>3</sup>, Suzanne B. Hodgkins<sup>4,5</sup>, Athan Basiliko<sup>6</sup>, Evan S. Kane<sup>3,7</sup>, Roxane Andersen<sup>8</sup>, Rebekka R. E. Artz<sup>9</sup>, Juan C. Benavides<sup>10</sup>, Brian W. Benscoter<sup>11,12</sup>, Werner Borken<sup>13</sup>, Luca Bragazza<sup>14</sup>, Stefani M. Brandt<sup>15</sup>, Suzanna L. Bräuer<sup>16</sup>, Michael A. Carson<sup>6</sup>, Dan Charman<sup>17</sup>, Xin Chen<sup>18</sup>, Beverley R. Clarkson<sup>19</sup>, Alexander R. Cobb<sup>20</sup>, Peter Convey<sup>21</sup>, Jonn del Águila Pasquel<sup>22</sup>, Andrea S. Enriquez<sup>23</sup>, Howard Griffiths<sup>24</sup>, Samantha P. Grover<sup>25</sup>, Charles F. Harvey<sup>26</sup>, Lorna I. Harris<sup>27,28</sup>, Christina Hazard<sup>29</sup>, Dominic Hodgson<sup>21</sup>, Alison M. Hoyt<sup>30</sup>, John Hribljan<sup>31</sup>, Jyrki Jauhiainen<sup>32,33</sup>, Sari Juutinen<sup>34</sup>, Klaus-Holger Knorr<sup>35</sup>, Randall K. Kolka<sup>36</sup>, Mari Könönen<sup>32,37</sup>, Tuula Larmola<sup>33</sup>, Carmody K. McCalley<sup>38</sup>, James McLaughlin<sup>39</sup>, Tim R. Moore<sup>28</sup>, Nadia Mykytczuk<sup>40</sup>, Anna E. Normand<sup>41</sup>, Virginia Rich<sup>5</sup>, Nigel Roulet<sup>28</sup>, Jessica Royles<sup>24</sup>, Jasmine Rutherford<sup>42</sup>, David S. Smith<sup>43</sup>, Mette M. Svenning<sup>44</sup>, Leho Tedersoo<sup>45,46</sup>, Pham Q. Thu<sup>47</sup>, Carl C. Trettin<sup>48</sup>, Eeva-Stiina Tuittila<sup>37</sup>, Zuzana Urbanová<sup>49</sup>, Ruth K. Varner<sup>50</sup>, Meng Wang<sup>51</sup>, Zheng Wang<sup>52</sup>, Matt Warren<sup>53</sup>, Magdalena M. Wiedermann<sup>54</sup>, Shanay Williams<sup>55</sup>, Joseph B. Yavitt<sup>56</sup>, Zhi-Guo Yu<sup>57</sup>, Zicheng Yu<sup>58,59</sup>, and Jeffrey P. Chanton<sup>1</sup>

<sup>1</sup>Department of Earth, Ocean, and Atmospheric Science, Florida State University, Tallahassee, FL, USA, <sup>2</sup>Department of Biology, Department of Environmental and Forest Biology, Syracuse University, State University of New York College of Environmental Science and Forestry, Syracuse, NY, USA, <sup>3</sup>USDA Forest Service, Houghton, MI, USA, <sup>4</sup>Department of Chemistry and Biochemistry, Florida State University, Tallahassee, FL, USA, <sup>5</sup>Department of Microbiology, The Ohio State University, Columbus, OH, USA, 6Department of Biology and the Vale Living with Lakes Centre, Laurentian University, Sudbury, ON, Canada, 7College of Forest Resources and Environmental Science, Michigan Technological University, Houghton, MI, USA, 8Environmental Research Institute, University of the Highlands and Islands, Inverness, Scotland, UK, <sup>9</sup>Ecological Sciences, James Hutton Institute, Aberdeen, UK, <sup>10</sup>Department of Ecology and Territory, Pontificia Universidad Javeriana, Bogota, Colombia, 11Department of Biological Sciences, Florida Atlantic University, Boca Raton, FL, USA, <sup>12</sup>Now at U.S. Department of Energy, Office of Science, Biological and Environmental Research, Washington, D.C., USA, <sup>13</sup>Soil Ecology, University of Bayreuth, Bayreuth, Germany, <sup>14</sup>Agroscope, Field-Crop Systems and Plant Nutrition, Research Division Plant Production Systems, Nyon, Switzerland, <sup>15</sup>Department of Biological Sciences, Humboldt State University, Arcata, CA, USA, <sup>16</sup>Department of Biology, Appalachian State University, Boone, NC, USA, <sup>17</sup>University of Exeter, Prince of Wales Road, Exeter, UK, <sup>18</sup>College of Life Sciences, Zhejiang University, Hangzhou, China, <sup>19</sup>Landcare Research New Zealand Ltd, Hamilton, New Zealand, 20Center for Environmental Sensing and Modeling, Singapore-MIT Alliance for Research and Technology, Singapore, Singapore, <sup>21</sup>British Antarctic Survey, Cambridge, UK, <sup>22</sup>Universidad Nacional de la Amazonia Peruana and Instituto de Investigaciones de la Amazonia Peruana, Iquitos, Peru, <sup>23</sup>Instituto de Investigaciones Forestales y Agropecuarias (CONICET-INTA), Río Negro, Argentina, 24 Department of Plant Sciences, University of Cambridge, Cambridge, UK, 25 Applied Chemistry and Environmental Science, RMIT University, Melbourne, VIC, Australia, <sup>26</sup>Massachusetts Institute of Technology and Singapore-MIT Alliance for Research and Technology, Cambridge, Massachusetts, USA, <sup>27</sup>Department of Renewable Resources, University of Alberta, Edmonton, AB, Canada, <sup>28</sup>Department of Geography, McGill University, Montreal, QC, Canada, <sup>29</sup>Environmental Microbial Genomics Group, Laboratoire Ampère, École Centrale de Lyon, Université de Lyon, Ecully, France, <sup>30</sup>Max Planck Institute for Biogeochemistry, Jena, Germany, <sup>31</sup>Department of Biology, University of Nebraska Omaha, Omaha, NE, USA, <sup>32</sup>Department of Forest Sciences, University of Helsinki, Helsinki, Finland, <sup>33</sup>Natural Resources Institute Finland (Luke), Helsinki, Finland, <sup>34</sup>Ecosystems and Environment Research Program, Faculty of Biological and Environmental Sciences, University of Helsinki, Helsinki, Finland, <sup>35</sup>Institute of Landscape Ecology, Ecohydrology & Biogeochemistry Group, University of Muenster, Muenster, Germany, <sup>36</sup>USDA Forest Service, Grand Rapids, MN, USA, <sup>37</sup>School of Forest Sciences, University of Eastern Finland, Joensuu, Kuopio, Finland, <sup>38</sup>Gosnell School of Life Sciences, Rochester Institute of Technology, Rochester, NY, USA, <sup>39</sup>Ontario Forest Research Institute, Sault Ste. Marie, ON, Canada, <sup>40</sup>Vale Living with Lakes Centre, Laurentian University, Sudbury, ON, Canada, <sup>41</sup>Soil and Water Sciences, University of Florida, Gainesville, FL, USA, <sup>42</sup>Department of Biodiversity, Conservation and Attractions, Kensington, W.A., Australia, <sup>43</sup>Department of Biology, California State University San Bernardino, San Bernardino, CA, USA, 44Department of Arctic and Marine Biology, UiT, The Arctic University of Norway, Tromsø, Norway, <sup>45</sup>Institute of Ecology and Earth Sciences, University of Tartu, Tartu, Estonia, <sup>46</sup>College of Science, King Saud University, Riyadh, Saudi Arabia, 47 Forest Protection Research Centre, Vietnamese Academy of Forest Sciences, Hanoi City, Vietnam, <sup>48</sup>USDA Forest Service, Southern Research Station, Cordesville, SC, USA, <sup>49</sup>Department of Ecosystem Biology, University of South Bohemia in České Budějovice, České Budějovice, Czech Republic, 50 Department of Earth Sciences and



Investigation: Brittany A. Verbeke, Louis J. Lamit, Erik A. Lilleskov, Suzanne B. Hodgkins, Nathan Basiliko, Evan S. Kane, Roxane Andersen, Rebekka R. E. Artz, Juan C. Benavides, Brian W. Benscoter, Werner Borken, Luca Bragazza, Stefani M. Brandt, Suzanna L. Bräuer, Michael A. Carson, Dan Charman, Xin Chen, Beverlev R. Clarkson, Alexander R. Cobb, Peter Convey, Jhon del Águila Pasquel, Andrea S. Enriquez, Howard Griffiths, Samantha P. Grover, Charles F. Harvey, Lorna I. Harris, Christina Hazard, Dominic Hodgson, Alison M. Hoyt, John Hribljan, Jvrki Jauhiainen, Sari Juutinen, Klaus-Holger Knorr, Randall K. Kolka, Mari Könönen, Tuula Larmola, Carmody K. McCalley, James McLaughlin, Tim R. Moore, Nadia Mykytczuk, Anna E. Normand, Virginia Rich, Nigel Roulet, Jessica Royles, Jasmine Rutherford, David S. Smith, Mette M. Svenning, Leho Tedersoo, Pham Q. Thu, Carl C. Trettin, Eeva-Stiina Tuittila, Zuzana Urbanová, Ruth K. Varner, Meng Wang, Zheng Wang, Matt Warren, Magdalena M. Wiedermann, Shanay Williams, Joseph B. Yavitt, Zhi-Guo Yu, Zicheng Yu, Jeffrey P Chanton

Methodology: Brittany A. Verbeke, Louis J. Lamit, Erik A. Lilleskov, Suzanne B. Hodgkins, Nathan Basiliko, Jeffrey P. Chanton

**Project Administration:** Louis J. Lamit, Erik A. Lilleskov, Suzanne B. Hodgkins, Jeffrev P. Chanton

Resources: Erik A. Lilleskov, Suzanne B. Hodgkins, Jeffrey P. Chanton Writing – original draft: Brittany A. Verbeke, Louis J. Lamit, Erik A.

A. Verbeke, Louis J. Lamit, Erik A. Lilleskov, Suzanne B. Hodgkins, Jeffrey P. Chanton

Writing - review & editing: Brittany A. Verbeke, Louis J. Lamit, Erik A. Lilleskov, Suzanne B. Hodgkins, Nathan Basiliko, Evan S. Kane, Roxane Andersen, Rebekka R. E. Artz, Juan C. Benavides, Brian W. Benscoter, Werner Borken, Luca Bragazza, Stefani M. Brandt, Suzanna L. Bräuer, Michael A. Carson, Dan Charman, Xin Chen, Beverley R. Clarkson, Alexander R. Cobb, Peter Convey, Jhon del Águila Pasquel, Andrea S. Enriquez, Howard Griffiths, Samantha P. Grover, Charles F. Harvey, Lorna I. Harris, Christina Hazard, Dominic Hodgson, Alison M. Hovt, John Hribljan, Jyrki Jauhiainen, Sari Juutinen, Klaus-Holger Knorr, Randall K. Kolka, Mari Könönen, Tuula Larmola, Carmody K. McCalley, James McLaughlin, Tim R. Moore, Nadia Mykytczuk, Virginia Rich, Nigel Roulet, Jessica Royles, Jasmine Rutherford, David S. Smith, Mette M. Svenning, Leho Tedersoo, Pham Q. Thu, Carl C. Trettin, Eeva-Stiina Tuittila, Zuzana Urbanová, Ruth K. Varner, Meng Wang, Zheng Wang, Matt Warren, Magdalena M. Wiedermann, Shanav Williams, Joseph B. Yavitt, Zhi-Guo Yu, Zicheng Yu, Jeffrey P. Chanton

the Study of Earth, Oceans and Space, University of New Hampshire, Durham, NH, USA, <sup>51</sup>State Environmental Protection Key Laboratory of Wetland Ecology and Vegetation Restoration, Institute for Peat and Mire Research, Northeast Normal University, Changchun, Jilin, China, <sup>52</sup>College of Forestry, Hebei Agricultural University, Baoding, China, <sup>53</sup>Earth Innovation Institute, San Francisco, CA, USA, <sup>54</sup>Department of Biological Sciences, University of Cincinnati, Cincinnati, OH, USA, <sup>55</sup>Department of Soil Science, College of Agriculture and Bioresources, University of Saskatchewan, Saskatoon, SK, Canada, <sup>56</sup>Department of Natural Resources & the Environment, Cornell University, Ithaca, NY, USA, <sup>57</sup>School of Hydrology and Water Resources, Nanjing University of Information Science and Technology, Nanjing, China, <sup>58</sup>Earth and Environmental Sciences, Lehigh University, Bethlehem, PA, USA, <sup>59</sup>Institute for Peat and Mire Research, School of Geographical Sciences, Northeast Normal University, Changchun, Jilin, China

**Abstract** Peatlands contain a significant fraction of global soil carbon, but how these reservoirs will respond to the changing climate is still relatively unknown. A global picture of the variations in peat organic matter chemistry will aid our ability to gauge peatland soil response to climate. The goal of this research is to test the hypotheses that (a) peat carbohydrate content, an indicator of soil organic matter reactivity, will increase with latitude and decrease with mean annual temperatures, (b) while peat aromatic content, an indicator of recalcitrance, will vary inversely, and (c) elevation will have a similar effect to latitude. We used Fourier Transform Infrared Spectroscopy to examine variations in the organic matter functional groups of 1034 peat samples collected from 10 to 20, 30–40, and 60–70 cm depths at 165 individual sites across a latitudinal gradient of 79°N–65°S and from elevations of 0–4,773 m. Carbohydrate contents of high latitude peat were significantly greater than peat originating near the equator, while aromatic content showed the opposite trend. For peat from similar latitudes but different elevations, the carbohydrate content was greater and aromatic content was lower at higher elevations. Higher carbohydrate content at higher latitudes indicates a greater potential for mineralization, whereas the chemical composition of low latitude peat is consistent with their apparent relative stability in the face of warmer temperatures. The combination of low carbohydrates and high aromatics at warmer locations near the equator suggests the mineralization of high latitude peat until reaching recalcitrance under a new temperature regime.

**Plain Language Summary** Peatlands are a large global soil carbon reservoir, containing a quantity of carbon that is equivalent to about half or more of the carbon dioxide in the atmosphere. What will be their fate on a warming planet? Across a latitudinal gradient from 79°N to 65°S, we measured the fraction of the peat made up of carbohydrates, which are easily decomposed, and of aromatics, which are less easily decomposed. We found that peat from high latitudes and high elevations had greater carbohydrate content, while aromatic content showed the opposite trend. Larger carbohydrate content in organic matter indicates greater decomposability, while greater aromatic content indicates lower decomposability. We suggest that this latitude/elevation difference that we observed predicts how high-latitude and high-elevation peats may change under warmer conditions. Our work indicates that while a large portion of the carbohydrate fraction in these peats could be lost upon warming, releasing greenhouse gases, a residual fraction will survive and become more aromatic-rich, making the remainder more resistant to rapid decomposition.

## 1. Introduction

Northern (boreal and subarctic) peatlands are estimated to contain  $500 \pm 100$  Gt of carbon (C), which accounts for a substantial fraction of global soil C, while only occupying about 3% of the terrestrial global surface (Gorham, 1991; Yu et al., 2010; Loisel et al., 2014, 2017), although a recent controversial estimate of peatland carbon is considerably greater (Nichols & Peteet, 2019). Tropical peatlands have only recently been included in the global soil carbon pool estimate, and even though northern peatlands contribute much more to this reservoir (estimates range from 455 to 621 Pg C; Gorham, 1991; Yu et al., 2010), peatlands closer to the equator add an estimated 65–105 Pg C (Dargie et al., 2017; Page et al., 2011; Yu et al., 2010).

Carbon accumulation measurements in peatland soils usually consist of defining the areal extent, depth, and carbon content of the peat, but the organic matter carbon quality should also be considered to accurately evaluate peatland soil carbon stability and potential carbon emissions in the face of climate change or other perturbations. Carbon accumulation and emission estimates could be improved by taking into account the chemical composition of the peat, which is one of the primary controls of degradability because of different activation energies of chemical bonds (Baldock et al., 1997; Hemingway et al., 2019; Keiluweit et al., 2016; LaRowe & Van Cappellen, 2011; Leifeld et al., 2012). One simple way to estimate organic matter carbon quality is to measure both carbohydrate

(O-alkyl carbon) content, which affects decomposition potential, and aromatic content, which affects recalcitrance (Baldock et al., 1997; Keiluweit et al., 2016).

#### 1.1. FTIR Organic Matter Functional Groups

Peat can be classified by the major components of organic matter functional groups, which have different biological and chemical availabilities with respect to mechanisms of decomposition and transformation (Normand, 2017; Tfaily et al., 2014). Fourier Transform Infrared Spectroscopy (FTIR) is an excellent tool to classify these groups based on the infrared absorption due to the vibrational properties of the atomic bonds. The major structural assignments of peaks from FTIR spectra (wavenumbers 4000-650 cm<sup>-1</sup> of the mid-IR range) are carbohydrates, aromatics, acids, and aliphatics. While the peak heights alone are not directly quantitative, after area normalization and baseline correction, comparing peak heights between samples is useful for determining the relative abundance of major structural groups (Hodgkins et al., 2018). In particular, the relative abundances of carbohydrates and aromatics are very well suited for evaluating levels of humification or decomposition. Carbohydrates (O-alkyl-C) are primarily found in peat-forming plants as cellulose, and are the most bio-labile constituent of peat (Hemingway et al., 2019; LaRowe & Van Cappellen, 2011; Tfaily et al., 2014). O-alkyl-C abundances have been shown to be the best proxy for decomposition potential compared to other factors such as soil pH, depth (age), and element ratios such as C/N, O/C, and H/C (Baldock et al., 1997; Leifeld et al., 2012). Aromatic compounds, including phenolics and some undissociated organic acids originating from plants like Sphagnum, are known to inhibit the decomposition of organic matter (Ballance et al., 2012; Bengtsson et al., 2018; Hájek et al., 2011; Hodgkins et al., 2014) and are much more refractory than carbohydrates. Lignin, a complex amorphous aromatic compound found in vascular peat-forming plants, has been identified as relatively stable in anaerobic peatland environments (Hodgkins et al., 2016; Kalbitz et al., 2003; Kirk & Farrell, 1987). Degradation of O-alkyl-C compounds by microbes can lead to residual enrichment of more recalcitrant components in peat (Tfaily et al., 2014). Peat decomposition is somewhat different from the decomposition of organic matter in mineral soils in that preservation through interaction with mineral grains (Hemingway et al., 2019) is not a factor in these mineral depleted soils.

#### 1.2. Evidence for a Global Gradient

Our previous research has evaluated peat stability over a global latitudinal gradient based on FTIR spectra but with a much smaller set of samples (Hodgkins et al., 2018). Carbohydrate and aromatic content of peat from seven different locations spanning the latitudes 4°N-68°N were determined based upon calibration curves between the FTIR spectra peak heights and a set of reference samples that had been subjected to wet chemistry measurements of the same organic matter functional groups to determine % cellulose + hemicellulose (carbohydrates) and % Klason lignin (aromatics). Hodgkins et al. (2018) conducted FTIR analyses on both peat cores and fresh plant materials from the locations and found that the estimated % carbohydrates increased from lower latitudes to higher latitudes, decreased with depth, while the estimated % aromatics showed the opposite trend. It was also inferred from the fresh plant material spectra that their initial carbohydrate and aromatic content could contribute to the variations in relative abundance of carbohydrates and aromatics in peat between latitudes.

#### 1.3. Botanical Composition of Peat

One key factor affecting peat chemistry is the chemical characteristics of the organic matter inputs (Zeh et al., 2020). Peatlands vary in vegetation composition, and the dominant factors affecting plant community composition include latitude, mean annual temperature (MAT), elevation, and hydrology, especially ground-water versus rainwater inputs. The resulting plant community variation affects the type of organic inputs and the degradation of organic matter (Wang et al., 2021). At higher northern latitudes where most peatlands are located, peat is often composed of non-woody litter. Graminoids typically dominate litter inputs at minerotrophic (groundwater-fed) peatlands, characterized by higher pH. Conversely, *Sphagnum* moss litter dominates in the more acidic ombrotrophic (rain-fed) and oligotrophic (nutrient-poor) northern peatlands (bogs and poor fens; Thomas & Pearce, 2004). Some *Sphagnum* peatlands are also present at lower latitudes, typically at higher elevations (Grover & Baldock, 2012; Urbina & Benavides, 2015). Graminoid litter inputs tend to be relatively labile, decomposing rapidly if exposed to oxygen (Scheffer et al., 2001; Verhoeven & Toth, 1995; Vitt et al., 2009). In contrast, *Sphagnum* mosses are more resistant to decomposition, and so are considered keystone species in

northern acidic peatland ecosystems (Van Breemen, 1995; Walker et al., 2017). *Sphagnum* mosses have unique physical and chemical properties that help preserve organic matter including microbiocidal pectins (Ballance et al., 2012), high cation exchange capacity (Bengtsson et al., 2018; Painter, 1991; Thomas & Pearce, 2004), and some lignin-like phenolics (Bengtsson et al., 2018).

At lower latitudes and elevations, peat swamp forests are the dominant peatland types (Gumbricht et al., 2017; Page et al., 2011). These are dominated by angiosperm trees and palms, and hence the dominant inputs to these peatlands are woody or palm litter with a relatively high lignin content (Dargie et al., 2017; Gandois et al., 2013; Könönen et al., 2016; Lahteenoja and Page, 2011; Lahteenoja et al., 2009). However, our measurements of *Shorea albida*, a tree dominant in tropical Southeast Asia, were roughly 40% carbohydrate and 30% aromatic as determined with FTIR (Hodgkins et al., 2018). Wood is relatively high in cellulose (Hodgkins et al., 2018). At higher northern latitudes, peat forming environments can, in some cases, be dominated by conifer species such as northern white cedar (*Thuja occidentalis*) at higher pH, or Pinaceae (e.g., *Larix, Picea*, and *Pinus*) at lower pH. Other rarer peatland types include cushion peatlands dominated by a diversity of herbaceous monocot and dicot cushion-forming species common in the Andes and Patagonia (Broder et al., 2012; Cooper et al., 2010; Mathijessen et al., 2019), restiad peatlands dominated by Restionacaeae (Poales) in Australia and New Zealand (Clarkson et al., 2017), and graminoid peatlands like the Florida Everglades.

#### 1.4. Peat Organic Matter Decomposition

In oxic soils, the depolymerization of organic matter is largely dependent on aerobic decomposition processes with oxygen as the terminal electron acceptor; whereas under oxygen-depleted conditions, oxidative enzyme activity is suppressed, anaerobic decomposers are dominant, and alternative electron acceptors are utilized along with hydrolytic enzymes (Keiluweit et al., 2016). Although hydrolytic enzymes can function under anoxic conditions, lower temperatures can reduce rates of enzyme activity, apparently by constraining enzymatic decomposition of polymers like cellulose (Conant et al., 2011). When organic matter decomposition is controlled by anaerobic decomposers, aromatics are selectively preserved (Baldock et al., 1997). Using solid-state <sup>13</sup>C Nuclear Magnetic Resonance spectroscopy, Baldock et al. (1997) showed many examples where the accumulation of aromatic carbon structures with decomposition was closely associated with a decrease in O-alkyl-C structures. They suggested that since an increase in aromatic content in humified peat coincided with carbohydrate loss, the relative abundances could serve as a sensitive index of decomposition. Biester et al. (2014) compared a number of decomposition proxies and found that the changes observed were mainly attributed to decomposition and to a lesser extent due to change in vegetation. They found preferential decomposition of phenols and polysaccharides and enrichment of aromatics with decomposition. Keiluweit et al. (2016) also showed that soil organic matter decomposition is constrained by the lability and recalcitrance of the carbon moieties present. Depolymerization of organic matter is limited by hydrolysis-resistant compounds such as lignin-like aromatics, whereas O-alkyl-C structures like carbohydrates are readily degradable by hydrolysis and fermentation (Keiluweit et al., 2016). Thus, humification of organic matter-and consequent decrease in O-alkyl-C and increase in aromatic-C-acts as a negative feedback to further decomposition, as the remaining organic matter has a lower content of labile O-alkyl structures and a higher content of recalcitrant aromatics. Low intensity fires can also affect organic matter degradation rates (Flanagan et al., 2020).

In addition to direct temperature effects on peat diagenesis, some latitudinal differences in peat chemistry could derive from the relative differences in the chemistry of peat litter inputs. Whatever its origin, there is likely a larger store of cellulose and other homopolymers that is preserved in colder high-latitude and high-elevation peat that would be susceptible to increased enzymatic depolymerization under a warmer climate (Hodgkins et al., 2018). With climate-change mediated temperature increase, which is relatively greater at high latitudes (Dorrepaal et al., 2009; Gallego-Sala et al., 2018; Meehl et al., 2007; Tingley & Huybers, 2013), a portion of the soil organic carbon that was formerly stable may become more accessible to microbial decomposition, via direct effect of rising temperature, indirect effects of water tables (Bäckstrand et al., 2010; Fenner & Freeman, 2011), or due to vegetation changes (McPartland et al., 2020; Norby et al., 2019). However, the timescale of this process is not known (Laine et al., 2019).

#### 1.5. Study Objectives

This study aimed to characterize the chemical composition of peat as defined by FTIR spectroscopy of organic matter functional groups. A total of 1335 peat samples were measured using FTIR. After omitting samples that had





Figure 1. Peatland sampling locations measured in this study and the corresponding mean annual temperatures (°C). Peat cores were collected across a global latitudinal gradient of  $79^{\circ}N-65^{\circ}S$ .

a high enough mineral content to create interference with the carbohydrate peak as described in the methods (Broder et al., 2012; Zimmermann et al., 2007), a total of 1034 samples across a latitudinal gradient of approximately 79°N–65°S were included in the final analysis (Table S1 in Supporting Information S1). All peat samples used in this study were collected from unmodified locations, which have not been directly subjected to experimental treatments or extensive impacts from human activities. The objective of this study was to test the following hypotheses: (a) carbohydrate content will increase along a latitudinal gradient from the equator toward the poles (35–8,765 km from the equator was sampled); (b) aromatic content will decrease along this gradient; and (c) elevation will have a similar effect to latitude, with higher carbohydrates and lower aromatics at high elevations relative to low elevations at similar latitudes (0–4,773 m was sampled). The relationships of carbohydrates and aromatics with latitude and elevation will primarily be due to strong changes in MAT along latitude and elevation gradients. These patterns are predicted to MAT, and they are primarily driven by decomposition (Hodgkins et al., 2018). This large database of samples not only expands upon the previously observed global latitudinal trends in peat chemistry (Hodgkins et al., 2018) with a much larger number of samples, it introduces elevation as another significant influence on peat organic matter carbon quality and provides a global-scale explanation of the observed patterns.

#### 2. Methods

#### 2.1. Peat Sampling

We assessed the chemical composition of organic matter in 1034 peat samples representing 165 unmodified locations spread across all continents from tropical to high latitude peatlands (Figure 1; Table S1 in Supporting Information S1). The samples were a subset of those from a larger study (The global peatland microbiome project, Lamit and Lilleskov et al., personal communication 2021) aimed at characterizing microbial communities and biogeochemistry in the Earth's peatlands. At each location, we collected one or more cores (most frequently three cores) using a variety of sampling methods, and material from the 10–20 cm, 30–40 cm, and 60–70 cm depth increments below the peat surface was saved for processing (in some cases only the shallower depths were collected). All samples represent 10 cm depth increments. However, the diameter of the cores varied from  $\sim$ 5 cm diameter Russian peat corers to approximately 10 cm diameter box corers. From each depth increment, a representative  $\sim$ 50 g of fresh peat was saved for processing. Although variation in core size could lead to some extraneous variation being added to the response variables we measured, we do not believe that variation at this small sampling scale would confound the overall interpretation of the peat chemistry.

Samples were frozen, chilled or freeze-dried and rapidly shipped to the USDA Forest Service, Northern Research Station, Houghton, MI, USA for further processing. We recorded the latitude, longitude, and elevation of





Figure 2. Example of peak finding using the *R* script with a Fourier Transform Infrared Spectroscopy spectrum of a Minnesota bog/poor fen 10–20 cm depth sample. The organic matter functional groups are shown with baseline correction of each peak and normalization to the integrated area of the spectrum.

coring locations, and scored the von Post humification index (Ekono, 1981; von Post, 1922), an ordinal index of decomposition degree (see Rydin & Jeglum, 2013), on moist peat from each depth increment. The von Post index ranges from 1 to 10, with 1 being the least decomposed and 10 being the most decomposed, and is based on characteristics such as the integrity of plant material in the peat, and the viscosity and color of what passes through the fingers when the peat is squeezed.

### 2.2. FTIR Analysis

Subsamples of each depth increment were freeze-dried or oven-dried (55°C), then ground to a fine powder with a SPEX SamplePrep 5100 Mixer/Mill ball grinder. A PerkinElmer Spectrum 100 FTIR spectrometer fitted with a CsI beam splitter and a deuterated triglycine sulfate detector was used to collect % transmittance spectra between wavenumbers 4,000–650 cm<sup>-1</sup>. The spectra were obtained by averaging four scans at a resolution of 4 cm<sup>-1</sup> after placing samples directly on an attenuated total reflectance (ATR) crystal and applying force so that the samples were held against the crystal. The Universal ATR accessory with a single-reflectance system was made from a zinc selenide/ diamond composite. After both ATR-correction and baseline-correction by the instrument software, all spectra were converted from transmittance to absorbance. The standard deviations of carbohydrate and aromatic carbon values were within 5% of the mean values when 4 replicate aliquots were run and scanned four times. That is, if a sample was found to be 30% carbohydrate, the analytical error on 4 aliquots that were each scanned 4 times was 1.5%.

#### 2.3. Peak Finding

A custom *R* script was used to determine the exact location of peaks in pre-defined regions of the FTIR spectra (https://github.com/shodgkins/FTIRbaselines; R Core Team, 2018). Peak regions were located individually for every sample (Figure 2) because variations in peat chemistry slightly shifted the baseline endpoints of the peaks. The minimum absorbance values within the expected region of the peak were used to define the baseline. Several important organic matter functional groups are known to fall between wavenumbers  $4,000-650 \text{ cm}^{-1}$  and can be categorized by an absorbance peak maximum.

Carbohydrates appear as a peak with absorbance maximum at approximately 1,030 cm<sup>-1</sup>, aromatics at 1,510 cm<sup>-1</sup>, and 1,630 cm<sup>-1</sup> (added together for this study), acids at 1,720 cm<sup>-1</sup>, and aliphatics at 2,850 cm<sup>-1</sup> and 2,920 cm<sup>-1</sup>. After finding the endpoints of the peak, absorbance values were then baseline-corrected by subtracting everything below the baseline, and the exact peak locations and peak heights were then determined by locating the maximum baseline-corrected absorbance value between the baseline endpoints. The *R* script calculates both peak height and area, and additionally normalizes the absorbance values by dividing them by the total integrated area of the spectrum.

To avoid interference from silicates, which produce a large peak that interferes with the carbohydrate peak  $(\sim 1,030 \text{ cm}^{-1})$ , samples that contained clay (determined based on the presence of a peak at 780 cm<sup>-1</sup> combined with a large peak at  $\sim 1,030 \text{ cm}^{-1}$ ) were excluded from analysis (Table S1 in Supporting Information S1). For the 780 cm<sup>-1</sup> peak, area-normalized absorbance greater than 0.00005 was used as the cutoff as the sample likely contained silicate minerals. After the peak heights were baseline-corrected, normalized, and checked for mineral interference, they were then converted into estimated % cellulose + hemicellulose (carbohydrates) and estimated % Klason lignin (aromatics; by weight) using the linear calibration equations generated in Hodgkins et al. (2018).

Carbohydrate calibration: y = 49204x - 1.7606,

where y = estimated % carbohydrates and x = carbohydrate peak height (~1,030 cm<sup>-1</sup>). Standard error of y estimate:  $\pm 9\%$ .

Aromatic calibration: y = 42332x + 7.3476,

where x = combined aromatic peak heights (~1,510 cm<sup>-1</sup> and ~1,630 cm<sup>-1</sup>). Standard error of y estimate: ±5%.

#### 2.4. Additional Variables

Four additional types of variables were derived to utilize in subsequent analyses. First, the package leaderCluster (Arnold, 2014) implemented in R 3.6.3 (R core team, 2020) was used to group peat samples into 1 and 50 km radius spatial clusters. Second, distance from the equator was calculated from latitude values using the geosphere package (Hijmans, 2019). Third, MAT was obtained from the 2.5 min WorldClim (version 2.1) data set (Fick & Hijmans, 2017) using latitude and longitude coordinates for each sample, using the packages sp. (Pebesma & Bivand, 2005) and raster (Hijmans, 2020). Because WorldClim does not include Antarctic climate records, MAT for the Antarctic sites were obtained from long-term records of the closest weather stations (Palmer Station: https:// doi.org/10.1016/j.quascirev.2017.10.033; Vernadsky Station: http://www.nerc-bas.ac.uk/icd/gjma/faraday.temps. html). Fourth, we created binary vegetation variables to represent three key components of the peat forming vegetation associated with coring locations, which were coded as 0 when below a given % cover threshold, or 1 when equal or above the threshold. These included total ground cover of Sphagnum (threshold = 50% cover), total overstory tree canopy cover (threshold = 50% cover), and total Cyperaceae cover (threshold = 35% cover). In most cases, these variables were obtained from percent cover data collected from 1 m<sup>2</sup> quadrats centered over each core location, although in some cases they were estimated from photos of the core locations. We reasoned that at levels above these thresholds each of the vegetation types would be more likely to be important contributors to the botanical composition of the peat, at least in the upper peat profile, however preliminary analyses with the cover thresholds adjusted up or downwards yielded equivalent interpretations of statistical results. The threshold for Cyperaceae cover was set lower than the other variables because the diffuse nature of their aboveground portions led Cyperaceae species to be less dense than the other variables even at locations where they comprised the vast majority of the plant community and were the known primary peat forming vegetation.

#### 2.5. Statistical Analysis

A suite of linear mixed models were run using R 3.6.3 to identify key predictors of carbohydrates, aromatics, and the von Post index. All mixed models were run with the package lmerTest (Kuznetsova et al., 2017), with F-tests obtained using the Kenward-Roger approximation, and the package partR2 (Stoffel et al., 2020) was used to obtain  $R^2$  values. The first set of models used the full data set to examine the relationship of carbohydrates, aromatics and, von Post index with distance from the equator. Each model included distance from the equator as a continuous independent variable, and the categorical fixed effect of depth in the peat profile (10–20, 30–40, and 60–70 cm). Terms for individual peat core, 1 km radius spatial clusters and 50 km radius spatial clusters were also included in these models as a series of explicitly nested random intercept effects to account for potential non-independence of samples collected from the same core or nearby locations. We used distance from the equator instead of latitude in models to allow for direct comparisons among regression coefficients. Second, we fit models for carbohydrates, aromatics and the von Post index using subsets of the data from a single individual depth at a time; these models included distance from the equator and the 1 and 50 km radius spatial clusters

as random effects. Third, to incorporate the potential effect of elevation, we fit models with the full data set containing distance from the equator, depth, elevation, and the random effects of core plus 1 and 50 km radius spatial clusters. Fourth, additional models were run with the full data sets using MAT (instead of distance from the equator and elevation) as a continuous independent variable, the categorical fixed effect of depth in the peat profile (10–20, 30–40, and 60–70 cm), and the random effects of peat core, 1 km radius spatial clusters and 50 km radius spatial clusters. Models with MAT did not include distance from the equator and elevation because these variables strongly covaried with MAT (Distance from the equator: partial r = -0.93, P < 0.0001; elevation: partial r = -0.76, P < 0.0001) and their inclusion could lead to issues with collinearity.

To account for the influence of vegetation on estimated carbohydrates, estimated aromatics, and the von Post index, we reran the three models described above that included the full data set. These updated models included a model with distance from the equator, another with distance from the equator and elevation, and a third with MAT, as the independent variables of interest. Each of these models also included the three binary vegetation covariates (i.e.,  $\geq 50\%$  Sphagnum cover,  $\geq 50\%$  tree canopy cover,  $\geq 35\%$  Cyperaceae cover), as well as the fixed effect of depth in the peat profile and the random effects of peat core, 1 km radius spatial clusters and 50 km radius spatial clusters. Data points from cores that did not have at least one of the three vegetation covariates above the given threshold were excluded from these models because they could have been dominated by a variety of unique non-target peat forming vegetation types (Table S3 in Supporting Information S1).

Lastly, we investigated the relationship of carbohydrates and aromatics with the von Post index. These linear mixed models contained the von Post index as a continuous independent variable, as well as the random effects of peat core, 1 km radius spatial clusters and 50 km radius spatial clusters.

## 3. Results

There were strong latitudinal and depth gradients in peat organic matter chemistry. The estimated % carbohydrates in peat found through FTIR analysis increased with greater distances from the equator, while the estimated % aromatics decreased (Table 1; Figure 3), therefore confirming our first two hypotheses. Estimate carbohydrate content also strongly decreased with increasing depth in the peat profile, whereas estimated aromatic content increased (Table 1; Figures 3a and 3b). While the larger slopes associated with depth highlighted the sharp average change in carbohydrates and aromatics over the ~0.5 m depth gradient, semi-partial  $R^2$  values indicated that the more gradual chemical changes associated with increasing distance from the equator explained a greater proportion of the variance (particularly for carbohydrates; Table 1). Importantly, models fit individually within each depth showed very similar results as the full model containing samples from all depths (Table S2 in Supporting Information S1). The importance of the latitudinal gradient is highlighted in the FTIR spectra of peat samples from depth 10–20 cm from both the Northern and Southern Hemisphere (Figure 4), where it is clear that the carbohydrate peak height (~1,030 cm<sup>-1</sup>) was much larger in peat from higher latitudes relative to peat from lower latitudes, and the aromatic peaks (1,510 cm<sup>-1</sup> and 1,630 cm<sup>-1</sup>) showed the opposite trend. Carbohydrate content increased with elevation while aromatic content decreased, confirming our third hypothesis. When elevation was added as another variable in the models with distance from the equator and sampling depth, elevation explained less of the variance in carbohydrates and aromatics than did distance from the equator but more than depth (Table 1). However, the slopes for the elevation effect were larger in magnitude than those of distance from the equator because the changes in carbohydrates and aromatics occurred over a much shorter distance (Table 1). Incorporating elevation into the models with distance from the equator and depth increased the total amount of variance explained by the main effects in the models by  $\sim$ 50% (carbohydrates) to 100% (aromatics) of the variance explained in models lacking elevation (Table 1). Adding elevation into the models also strengthened the relationship (semi-partial  $R^2$  and slopes) between distance from the equator and the FTIR variables (Table 1). The patterns seen with the full data set were also evident within individual sampling depths (Table S2 in Supporting Information S1). When the peat samples were divided into groups of sampling locations below and above 1,500 m elevation (Figures 3c and 3d), the carbohydrate and aromatic content of peat from higher elevations more closely matched that of peat from higher latitudes even when the peat was from lower latitudes (Figure 5).

Mean annual temperature was a very strong predictor of carbohydrate and aromatic content. When MAT was used in place of distance from the equator and elevation in models, its explanatory power was similar to distance from the equator and elevation combined (Table 1; Figures 3e and 3f). For example, MAT and depth together



#### Table 1

Linear Mixed Model Results of Carbohydrate, Aromatic and von Post Responses to Distance From the Equator (Dist), Sampling Depth, Elevation (El), Mean Annual Temperature (MAT)<sup>a,b,c,d,e</sup>

Models	Distance from the equator $F$ ( <i>df</i> ) $P R^2 B$	Depth $F(df) P R^2 B$	Elevation $F(df) P R^2 B$	Mean annual temperature $F$ ( <i>df</i> ) $P R^2 B$	Full model R <sup>2</sup> (marginal/ conditional)
Carbohydrates					
Dist + Depth	24.7 (1, 62.7) <b>&lt;0.001</b> , 0.15, 0.0023	195.7 (2, 647.9) <b>&lt;0.001</b> , 0.08, -3.9/-7.8			0.22/0.84
Dist + El + Depth	56.6 (1, 53.8) <b>&lt;0.001</b> , 0.28, 0.0033	195.5 (2, 648.7) <b>&lt;0.001</b> , 0.07, -3.9/-7.8	23.9 (1, 57.5) <b>&lt;0.001</b> , 0.14, 4.6		0.36/0.85
MAT + Depth		196.4 (2, 649.2) <b>&lt;0.001</b> , 0.07, -3.9/-7.8		66.8 (1, 53.3) <b>&lt;0.001</b> , 0.33, -0.76	0.39/0.85
Aromatics					
Dist + Depth	13.2 (1, 65.3) <b>&lt;0.001</b> , 0.09, -0.00040	78.3 (2, 670.3) <b>&lt;0.001</b> , 0.06, 0.82/1.5			0.14/0.70
Dist + El + Depth	40.5 (1, 59.3) <b>&lt;0.001</b> , 0.22, -0.00065	78.2 (2, 671.8) <b>&lt;0.001</b> , 0.05, 0.81/1.5	28.5 (1, 63.7) <b>&lt;0.001</b> , 0.16, -1.2		0.29/0.70
MAT + Depth		78.5 (2, 671.2) <b>&lt;0.001</b> , 0.05, 0.82/1.5		27.0 (1, 60.2) <b>&lt;0.001</b> , 0.18, 0.13	0.23/0.71
von Post index					
Dist + Depth	11.8 (1, 63.5) <b>0.001</b> , 0.08, -0.00026	213.2 (2, 658.0) <b>&lt;0.001</b> , 0.13, 0.93/1.6			0.19/0.74
Dist + El + Depth	13.4 (1, 60.1) <b>0.001</b> , 0.10, -0.00032	212.8 (2, 657.9) <b>&lt;0.001</b> , 0.12, 0.92/1.6	1.6 (1, 64.5) 0.206, 0.02, -0.23		0.21/0.75
MAT + Depth		212.8 (2, 658.2) <b>&lt;0.001</b> , 0.12, 0.92/1.6		13.6 (1, 59.0) <b>&lt;0.001</b> , 0.10, 0.07	0.22/0.75

<sup>a</sup>All models also include three nested random effects terms that group samples by core, 1 km radius spatial clusters and 50 km radius spatial clusters. No null hypothesis test was applied to these factors. Bolded numbers indicate *p* values.  ${}^{b}R^{2}$  for individual main effects (fixed factors and numerical independent variables) represent semipartial  $R^{2}$  values. For full model  $R^{2}$ , marginal values are the proportion of variance explained by main effects, while conditional values represent the variance explained by main plus random effects.  ${}^{c}B$  = non-standardized regression coefficients. Two coefficients are reported for depth: the slope from the 10–20 to 30–40 cm depth, and then 10–20 to the 60–70 cm depth. Units for distance from the equator (Dist) and elevation (El) are kilometers. Units for mean annual temperature (MAT) are degrees Celsius. <sup>d</sup>For the carbohydrate and aromatic models in this table: Samples = 1,034, Core = 423; Clusters 1 km radius = 116; Clusters 50 km radius = 67. <sup>c</sup>For the von Post model in this table: Samples = 117, Clusters 1 km radius = 114; Clusters 50 km radius, = 65. This is because a few samples did not have von Post scores performed on them.

explained nearly four-tenths of the total variance in carbohydrates, while distance from the equator, elevation, and depth together explained slightly less (Table 1). The large contribution of MAT as a predictor for peat carbohydrate and aromatic content in the models, combined with its strong correlation with distance from the equator and elevation, suggest that the influence of distance from the equator and elevation on the relative abundances of carbohydrates and aromatics is primarily due to MAT.

In models that included the dominant peat forming surface vegetation as covariates, much more variation in carbohydrates, aromatics, and von Post was attributable to distance from the equator, elevation, MAT and/or depth than could be attributed to any of the vegetation covariates (Table S3 in Supporting Information S1). However, carbohydrates did show a distinct increase when *Sphagnum* was  $\geq$ 50%, and a distinct decline when overstory canopy cover was  $\geq$ 50% (Table S3 in Supporting Information S1), while the Cyperaceae covariate was not significant. In contrast, aromatics, and von Post did not show clear relationships with any of the vegetation covariates (Table S3 in Supporting Information S1). These results suggest that the relationships of carbohydrates, aromatics, and von Post with distance from the equator, elevation, and MAT are in part decoupled from shifts in surface peat forming vegetation.

Values of the von Post index shifted with distance from the equator, elevation, MAT, and depth (Table 1), and also covaried with the FTIR variables. The main effects in the models containing distance from the equator, elevation, depth and/or MAT all explained  $\sim$  one-fifth of the variance in the von Post index (Table 1), with the von Post



# **Global Biogeochemical Cycles**



**Figure 3.** Estimated % carbohydrates ( $\pm$ 9%) and aromatics ( $\pm$ 5%) for all 1034 peat samples with corresponding distance from the equator (km north or south). Peat samples are divided by depth categories 10–20 cm, 30–40 cm, and 60–70 cm for (a) carbohydrate content and (b) aromatic content, by elevation categories 0–1,500 m and 1500 + m for (c) carbohydrate content and (d) aromatic content, and by mean annual temperature (°C) for (e) carbohydrate content and (f) aromatic content. Statistical analysis presented in Table 1 and Table S2 in Supporting Information S1.

index decreasing with greater distance from the equator and elevation, and increasing with greater depth in the peat profile and MAT (Table 1, Table S2 in Supporting Information S1). The von Post index exhibited significant relationships with both FTIR variables (Figure 6), although von Post index was a stronger predictor of carbohydrates ( $F_{1,901.5} = 382.1$ , P < 0.001, B = -2.75, marginal  $R^2 = 0.21$ ) than aromatics ( $F_{1,1001.1} = 66.3$ , P < 0.001, B = 0.35, marginal  $R^2 = 0.07$ ). Importantly, higher von Post index values signify more decomposed peat, which is demonstrated by the strong increase in von Post index with depth (see also Figure S1 in Supporting Information S1); this relationship of von Post index with carbohydrates and aromatics confirms that these FTIR-derived chemical variables are also indicators of the degree of decomposition.

Comparisons of marginal to conditional full model  $R^2$  values indicate that peat chemistry exhibited strong spatial effects (Table 1; Tables S2 and S3 in Supporting Information S1). In all analyses, the total proportion of variance explained by random effects (individual peat core, 1 km radius spatial clusters and 50 km radius spatial clusters considered together) was typically greater than that explained by fixed effects, often by far more than double. This indicates that carbohydrates, aromatics, and von Post values are on average more similar for samples collected near each other than samples from distant locations.





**Figure 4.** Fourier Transform Infrared Spectroscopy spectra of 10–20 cm depth peat samples. (a) Locations in the Northern Hemisphere from low to high latitudes: Mendaram, Brunei (4.360°N); Mekong delta, Vietnam (9.219°N); Everglades, Florida, USA (25.869°N); Alligator River, North Carolina, USA (35.786°N); Dong Tu, NE China (42.271°N); Marcell, Minnesota, USA (47.514°N); Kuujjuarapik, Quebec, Canada (55.348°N); Copper River Delta, Alaska, USA (60.402°N); Stordalen Mire, Sweden (68.353°N). (b) Locations in the Southern Hemisphere from low to high latitudes: Sabangau Forest, Indonesia (2.192°S); Peruvian Amazon, Peru (4.691°S); Saunders Spring, Mandora Marsh (Walyarta) Great Sandy Desert, NW Australia (19.783°S); Poorginup Swamp, SW Australia (34.547°S); Puerto Blest, Argentina (41.041°S); South Island, New Zealand (43.055°S); Seno Skyring Patagonia, Chile (52.515°S); and Litchfield Island, Antarctic Peninsula (64.770°S). Carbohydrate peak is at 1,030 cm<sup>-1</sup>, and aromatic peaks are at 1,510 cm<sup>-1</sup> and 1,630 cm<sup>-1</sup>.

## 4. Discussion

We found that the estimated relative abundances of carbohydrates increased from the equator toward the poles, and the estimated relative abundances of aromatics decreased, supporting hypotheses 1 and 2. Higher elevation peat from lower latitude peatlands also had greater carbohydrate content and lower aromatic content than lower elevation peat from similar latitudes, supporting hypothesis 3. These differences in peat organic matter carbon quality may provide a sensitive proxy for decomposition potential, with higher relative abundances of





**Figure 5.** Fourier Transform Infrared Spectroscopy spectra of 10-20 cm depth peat samples from similar latitudes but different elevations. Temperate latitude samples are from Chattanooga Fen, Tennessee (37.866°N, 3,114 m elevation) and Sallie's Fen, New Hampshire (43.126°N, 120 m elevation). Tropical latitude samples are from Chingaza National Park, Colombia (4.678°N, 3,623 m elevation) and the Peruvian Amazon, Peru (4.692°S, 97 m elevation). Carbohydrate peak is at 1,030 cm<sup>-1</sup> and aromatic peaks are at 1,510 cm<sup>-1</sup> and 1,630 cm<sup>-1</sup>.

carbohydrates indicating bio-lability and higher relative abundances of aromatics indicating more recalcitrance (Baldock et al., 1997; Grover & Baldock, 2010, 2012; Treat et al., 2014). Organic matter with more carbohydrates would also have a higher nominal oxidation state of carbon (Keiluweit et al., 2016; LaRowe & Van Cappellen, 2011) offering a theoretical basis for greater reactivity of carbohydrates under anaerobic conditions. Peat with lower carbohydrate content and higher aromatic content was generally classified as more highly decomposed on the von Post index of humification. The less decomposed peat from higher latitudes showed greater potential for mineralization owing to the larger relative abundance of carbohydrates to aromatics, a trait shared by peat originating from higher elevations.

Peat stability and decomposition rates are influenced by a combination of many factors, including water table depth, electron acceptor availability (Blodau, 2011; Heimann et al., 2009), and microbial community (Bragazza et al., 2016; Hodgkins et al., 2015; LaRowe & Van Cappellen, 2011; Norby et al., 2019), but also by the source vegetation and associated organic matter quality (Wang et al., 2021; Zeh et al., 2020). After accounting for dominant peat-forming surface vegetation in our analyses, latitude and elevation had by far the strongest relationships



**Figure 6.** (a) Estimated % carbohydrates ( $\pm$ 9%) and (b) aromatics ( $\pm$ 5%) with corresponding von Post index values for all three (10–20, 30–40, and 60–70 cm) depths. The von Post index is a scale of humification, with 1 being the least decomposed and 10 being the most decomposed. Color scale is mean annual temperature (°C). Individual depth segments are plotted in Figure S1 in Supporting Information S1.

with carbohydrates and aromatics (Table S3 in Supporting Information S1). This is consistent with our hypothesis that the observed carbohydrate and aromatic patterns arose for the most part as a response to climate (Artz et al., 2008), and is not strongly driven by botanical composition at the global scale. However, we do recognize that there are limitations to our approach. Most importantly, surface vegetation does not necessarily reflect the botanical origin of the peat, especially for deeper depths in the profile that may have formed when the peatland was dominated by different plant species. An important next-step for the scientific community would be to utilize macrofossils to compare FTIR derived carbohydrate and aromatic values to the specific vegetation composition of the peat in each sample, but that is beyond the scope of the current study. Although not part of our explicit hypotheses, the analyses also show that a large portion of the variation in peat carbohydrates and aromatics can be attributed to random effects associated with core and spatial clusters. This strong spatial effect is to be expected because samples from nearby locations have very similar ecological characteristics, including climates, hydrology, and botany, that should similarly influence peat stability and decomposition. An important aim of this study was to decouple the global pattern from local spatial effects in order to pull out the global-scale gradient, and the fact that we found clear patterns in spite of the expected strong spatial effects adds to the robustness of the findings. There is likely both a direct climatic effect, and an indirect effect of climate on peat organic matter chemistry as mediated by changes in the dominant peatland types and their associated plant functional groups. The combination of the inhibitory effects of aromatic compounds from plants such as *Sphagnum* and shrubs (Wang et al., 2015), and the higher recalcitrance of lignin-like compounds, make aromatics one of the most influential factors in peat stability. Aromatic compounds from *Sphagnum* and vegetation with similar chemical composition found at higher latitudes are significant for peat stability in northern peatlands because the plant tissue decomposition rates are much slower than that of other source vegetation (Könönen et al., 2016; Moore et al., 2007; Scanlon & Moore, 2000), but the higher initial aromatic content of source vegetation at lower latitudes shown in Hodgkins et al. (2018) could increase the recalcitrance of the soil organic matter even without inhibitory compounds from *Sphagnum* (Könönen et al., 2015, 2018). Our FTIR-derived aromatics have been calibrated against "Klason Lignins" (Hodgkins et al., 2018), a quite diverse operationally defined pool that includes not just structural lignin, but also other aromatics such as tannins, and other non-lignin-derived polyphenols (De la Cruz et al., 2016).

Our finding of lower carbohydrates in peat near the equator is important because O-alkyl-C compounds have a higher carbon oxidation state than aromatic compounds, causing faster rates of decomposition under anaerobic conditions in peat with larger carbohydrate abundances (LaRowe & Van Cappellen, 2011). Higher aromatic abundances accompanied by lower carbohydrate abundances produce a reduced oxidation state, suggesting that peatlands exhibiting these chemical qualities will decompose at slower rates even with warmer temperatures (Hodgkins et al., 2018; Worrall et al., 2017). In contrast, decomposition rates are more rapid for surface peat than deep peat due to a higher concentration of bio-labile organic matter (Charman et al., 2013; Normand, 2017). Hodgkins et al. (2018) also suggested that warmer temperatures at lower latitudes could be causing surface peat to transition to a more recalcitrant state faster than at higher latitudes.

It is worth noting that based on von Post, the variation from almost undecomposed to completely decomposed peat (1–9) resulted in a 50% decrease in carbohydrates (42%–22%), whereas it resulted in only a 12% increase (24%–27%) in aromatics (Figure 6). Hodgkins et al. (2018) observed a relatively greater decrease in carbohydrates with depth while the increase in aromatics was less. These observations suggest that during peat decomposition, carbohydrates are lost at a faster rate than overall C, whereas aromatics appear to degrade at a slower rate than overall C. Our finding of increased humification with depth, as reflected in von Post index values as well as carbohydrate and aromatic content, is consistent with broad understanding of peat diagenesis. For example, based on these values, Scanlon and Moore (2000) showed that the level of humification generally increased with depth, and Charman et al. (2013) found that decomposition rates were slower at depth in anoxic zones where recalcitrant organic matter accumulates. Colder temperatures and anaerobic conditions are essential to peat preservation, but increased recalcitrance of soil organic matter at depth and at lower latitudes suggests that chemical quality might have a greater impact on preservation than previously thought. Higher abundance of O-alkyl carbon increases rates of C mineralization in peat soils (Grover & Baldock, 2010, 2012; Normand, 2017), so whatever the cause, there is evidence that high latitude and high elevation soils are poised for higher reaction rates than equatorial peats.

O-Alkyl-C compounds have been preserved in higher latitude peatlands due to colder temperatures, but with warmer temperatures (as predicted particularly for these regions), carbohydrates would likely be the first constituent of peat to be lost through decomposition processes while the aromatics would be left behind. Carbohydrate abundances are low in deeper peat (Hodgkins et al., 2018; Tfaily et al., 2014), so it is likely that mainly surface peat will be affected while deeper peat would not decompose as much (Wilson et al., 2016; Worrall et al., 2018). Such effects were observed in long-term drained peatlands–once more available compounds (as measured by FTIR) were depleted following a few decades of drainage, the decomposition slowed due to residual high content of recalcitrant compounds (Urbanová et al., 2018). Of course, it may be argued that recalcitrant compounds are not residuals, but rather are the products of decomposition itself (Prescott, 2010). We think of humification (Lehmann & Kleber, 2015) as the loss of oxygenated compounds from the peat, which is coincident with the increase of alkyl and aromatic residuals or products.

If our conceptual model derived from this gradient analysis is correct, namely that high carbohydrate containing peat is more labile in the face of rising temperatures, then in high-latitude peatlands subject to warming we speculate that the following may occur: initial increases in decomposition of more labile carbohydrate-rich surface peat, with preferential removals of carbohydrates, leading to increases in peat aromatic:carbohydrate ratios, which will in turn slow decomposition and constrain the resultant positive climate feedback of greenhouse gas Acknowledgments

Funding was provided by the USDA

Climate Change Program, the US Na-

DEB-1146149) to E.S. Kane and E.A.

Lilleskov. This study was funded in part

by the Office of Biological and Environ-

mental Research, Terrestrial Ecosystem

DOE contracts DE-SC0007144 and DE-

SC0012088. We also acknowledge fund-

ing from the National Science Foundation

for the EMERGE Biology Integration

For sample collection we thank Ian A.

University of Canterbury, Christchurch,

New Zealand; the Cass Field Station,

University of Canterbury: Geoff Zahn

of Utah Valley University, Orem, UT,

Oak Ridge, TN, USA; and Mark. P.

USA; Christopher W. Schadt, Biosciences

Division, Oak Ridge National Laboratory,

Waldrop, Geology, Minerals, Energy, and

Geophysics Science Center, USGS Menlo

mous reviewers for helpful comments on

Park, CA, USA. We thank two anony-

the manuscript.

Dickie and the Biological Sciences,

Institute, NSF Award # 2022070.

Science Program, under United States

Forest Service Northern Research Station

tional Science Foundation (grant number

emissions. Models that can capture this dynamic are more likely to accurately reflect the response of peatlands to climate change. However, further studies are necessary in order to better relate the expected chemical changes of stored peat to associated rates of greenhouse gas release in response to climate warming. It has been suggested that as the Earth transitioned from glacial times to an interglacial state during the Holocene, the soil organic reservoir transitioned from permafrost mineral soils to peat (Lindgren et al., 2018). It is our hope that our study has shed some light on what may occur as we now transition from the Holocene to the Anthropocene. We realize that a myriad of other factors in addition to temperature will affect peatland carbon storage including water table changes (Roulet et al., 1993), fires (Flanagan et al., 2020; Turetsky et al., 2015), and vegetation changes such as increasing shrubification and *Sphagnum* loss (Malhotra et al., 2020; McPartland et al., 2020; Norby et al., 2019). In addition, peat may be buried under mineral soils with changing climate and thus preserved for periods of time on the order of hundreds of thousands of years (Treat et al., 2019). Our study indicates that loss of carbohydrates or o-alkyl functionality plays a central role in long term carbon storage.

## **Data Availability Statement**

The data used in this analysis are publicly available at the US Forest Service Research Data Archive (at the following DOI: https://doi.org/10.2737/RDS-2021-0045).

#### References

- Arnold, T. B. (2014). leaderCluster: Leader clustering algorithm. R package version 1.2. Retrieved from https://CRAN.R-project.org/ package=leaderCluster
- Artz, R. R. E., Chapman, S. J., Jean Robertson, A. H., Potts, J. M., Laggoun-Défarge, F., Gogo, S., et al. (2008). FTIR spectroscopy can be used as a screening tool for organic matter quality in regenerating cutover peatlands. *Soil Biology and Biochemistry*, 40(2), 515–527. https://doi. org/10.1016/j.soilbio.2007.09.019
- Bäckstrand, K., Crill, P. M., Jackowicz-Korczynski, M., Mastepanov, M., Christensen, T. R., & Bastviken, D. (2010). Annual carbon gas budget for a subarctic peatland, Northern Sweden. *Biogeosciences*, 7, 95–108.
- Baldock, J. A., Oades, J. M., Nelson, P. N., Skene, T. M., Golchin, A., & Clarke, P. (1997). Assessing the extent of decomposition of natural organic materials using solid-state <sup>13</sup>C NMR spectroscopy. Australian Journal of Soil Research, 35, 1061–1083.
- Ballance, S., Kristiansen, K. A., Skogaker, N. T., Tvedt, K. E., & Christensen, B. E. (2012). The localisation of pectin in *Sphagnum* moss leaves and its role in preservation. *Carbohydrate Polymers*, 87, 1326–1331. https://doi.org/10.1016/j.carbpol.2011.09.020
- Bengtsson, F., Rydin, H., & Hájek, T. (2018). Biochemical determinants of litter quality in 15 species of *Sphagnum*. *Plant and Soil*, 425, 161–176. https://doi.org/10.1007/s11104-018-3579-8
- Biester, H., Knorr, K. H., Schellekens, J., Basler, A., & Hermanns, Y. M. (2014). Comparison of different methods to determine the degree of peat decomposition in peat bogs. *Biogeosciences*, 11, 2691–2707. https://doi.org/10.5194/bg-11-2691-2014
- Blodau, C. (2011). Thermodynamic control on terminal electron transfer and methanogenesis. Aquatic Redox Chemistry, 4, 65–83. https://doi.org/10.1021/bk-2011-1071.ch004
- Bragazza, L., Buttler, A., Robroek, B. J. M., Albrecht, R., Zaccone, C., Jassey, V. E. J., & Signarbieux, C. (2016). Persistent high temperature and low precipitation reduce peat carbon accumulation. *Global Change Biology*, 22, 4114–4123. https://doi.org/10.1111/gcb.13319
- Broder, T., Blodau, C., Biester, H., & Knorr, K. H. (2012). Peat decomposition records in three pristine ombrotrophic bogs in southern Patagonia. *Biogeosciences*, 9, 1479–1491. https://doi.org/10.5194/bg-9-1479-2012
- Charman, D. J., Beilman, D. W., Blaauw, M., Booth, R. K., Brewer, S., Chambers, F. M., et al. (2013). Climate-related changes in peatland carbon accumulation during the last millennium. *Biogeosciences*, 10, 929–944. https://doi.org/10.5194/bg-10-929-2013
- Clarkson, B., Whinam, J., Good, R., & Watts, C. (2017). Restoration of Sphagnum and restiad peatlands in Australia and New Zealand reveals similar approaches. *Restoration Ecology*, 25(2), 301–311. https://doi.org/10.1111/rec.12466
- Conant, R. T., Ryan, M. G., Ågren, G. I., Birge, H. E., Davidson, E. A., Eliasson, P. E., et al. (2011). Temperature and soil organic matter decomposition rates–Synthesis of current knowledge and a way forward. *Global Change Biology*, *17*(11), 3392–3404. https://doi. org/10.1111/j.1365-2486.2011.02496.x
- Cooper, D. J., Wolf, E. C., Colson, C., Vering, W., Granda, A., & Meyer, M. (2010). Alpine peatlands of the Andes, Cajamarca, Peru. Arctic Antarctic and Alpine Research, 42(1), 19–33. https://doi.org/10.1657/1938-4246-42.1.19
- Dargie, G. C., Lewis, S. L., Lawson, I. T., Mitchard, E. T. A., Page, S. E., Bocko, Y. E., & Ifo, S. A. (2017). Age, extent and carbon storage of the central Congo Basin peatland complex. *Nature*, 542, 86–90. https://doi.org/10.1038/nature21048
- De la Cruz, F. B., Osborne, J., & Barlaz, M. A. (2016). Determination of sources of organic matter in solid waste by analysis of phenolic copper oxide oxidation products of lignin. *Journal of Environmental Engineering*, 142(2), 04015076. https://doi.org/10.1061/(ASCE) EE.1943-7870.0001038
- Dorrepaal, E., Toet, S., van Logtestijn, R. S. P., Swart, E., van de Weg, M. J., Callaghan, T. V., & Aerts, R. (2009). Carbon respiration from subsurface peat accelerated by climate warming in the subarctic. *Nature*, 460(7255), 616–619. https://doi.org/10.1038/nature08216

Ekono. (1981). Report on energy use of peat, Contribution to U.N. Conference (p. 42). New Renewable Sources of Energy.

Fenner, N., & Freeman, C. (2011). Drought-induced carbon loss in peatlands. *Nature Geoscience*, 4, 895–900. https://doi.org/10.1038/ngeo1323
Fick, S. E., & Hijmans, R. J. (2017). WorldClim 2: New 1km spatial resolution climate surfaces for global land areas. *International Journal of Climatology*, 37(12), 4302–4315.

- Flanagan, N. E., Wang, H., Winton, S., & Richardson, C. J. (2020). Low-severity fire as a mechanism of organic matter protection in global peatlans: Thermal alteration slows decomposition. *Global Change Biology*, 26(7), 3930–3946.
- Gallego-Sala, A. V., Charman, D. J., Brewer, S., Page, S. E., Prentice, I. C., Friedlingstein, P., et al. (2018). Latitudinal limits to the predicted increase of the peatland carbon sink with warming. *Nature Climate Change*, 8, 907–913. https://doi.org/10.1038/s41558-018-0271-1

- Gandois, L., Cobb, A. R., Hei, I. C., Lim, L. B. L., Salim, K. A., & Harvey, C. F. (2013). Impact of deforestation on solid and dissolved organic matter characteristics of tropical peat forests: Implications for carbon release. *Biogeochemistry*, 114(1–3), 183–199.
- Gorham, E. (1991). Northern peatlands: Role in the carbon cycle and probable responses to climatic warming. *Ecological Applications*, 1(2), 182–195.
- Grover, S. P. P., & Baldock, J. A. (2010). Carbon decomposition processes in a peat from the Australian Alps. *European Journal of Soil Science*, 61(2), 217–230. https://doi.org/10.1111/j.1365-2389.2009.01220.x
- Grover, S. P. P., & Baldock, J. A. (2012). Carbon chemistry and mineralization of peat soils from the Australian Alps. *European Journal of Soil Science*, 63(2), 129–140.
- Gumbricht, T., Roman-Cuesta, R. M., Verchot, L., Herold, M., Wittmann, F., Householder, E., et al. (2017). An expert system model for mapping tropical wetlands and peatlands reveals South America as the largest contributor. *Global Change Biology*, 23(9), 3581–3599. https://doi. org/10.1111/gcb.13689
- Hájek, T., Ballance, S., Limpens, J., Zijlstra, M., & Verhoeven, J. T. A. (2011). Cell-wall polysaccharides play an important role in decay resistance of *Sphagnum* and actively depressed decomposition in vitro. *Biogeochemistry*, 103, 45–57. https://doi.org/10.1007/s10533-010-9444-3
- Heimann, A., Jakobsen, R., & Blodau, C. (2009). Energetic constraints on H<sub>2</sub>-dependent terminal electron accepting processes in anoxic environments: A review of observations and model approaches. *Environmental Science and Technology*, 44(1), 24–33. https://doi.org/10.1021/es9018207
- Hemingway, J. D., Rothman, D. H., Grant, K. E., Rosengard, S. Z., Eglinton, T. I., Derry, L. A., & Galy, V. V. (2019). Mineral protection regulates long-term global preservation of natural organic carbon. *Nature*, 570(7760), 228–231. https://doi.org/10.1038/s41586-019-1280-6
- Hijmans, R. J. (2019). Geosphere: Spherical Trigonometry. R package version 1.5-10. Retrieved from https://CRAN.R-project.org/ package=geosphere
- Hijmans, R. J. (2020). Raster: Geographic data analysis and modeling. R package version 3.4-5. Retrieved from https://CRAN.R-project.org/ package=raster
- Hodgkins, S. B., Chanton, J. P., Langford, L. C., McCalley, C. K., Saleska, S. R., Rich, V. I., et al. (2015). Soil incubations reproduce field methane dynamics in a subarctic wetland. *Biogeochemistry*, 126(1–2), 241–249. https://doi.org/10.1007/s10533-015-0142-z
- Hodgkins, S. B., Richardson, C. J., Dommain, R., Wang, H., Glaser, P. H., Verbeke, B. A., et al. (2018). Tropical peatland carbon storage linked to global latitudinal trends in peat recalcitrance. *Nature Communications*, 9, 3640. https://doi.org/10.1038/s41467-018-06050-2
- Hodgkins, S. B., Tfaily, M. M., McCalley, C. K., Logan, T. A., Crill, P. M., Saleska, S. R., et al. (2014). Changes in peat chemistry associated with permafrost thaw increase greenhouse gas production. *Proceedings of the National Academy of Sciences of the United States of America*, 111(16), 5819–5824. https://doi.org/10.1073/pnas.1314641111
- Hodgkins, S. B., Tfaily, M. M., Podgorski, D. C., McCalley, C. K., Saleska, S. R., Crill, P. M., et al. (2016). Elemental composition and optical properties reveal changes in dissolved organic matter along a permafrost thaw chronosequence in a subarctic peatland. *Geochimica et Cosmochimica Acta*, 187, 123–140. https://doi.org/10.1016/j.gca.2016.05.015
- Kalbitz, K., Schmerwitz, J., Schwesig, D., & Matzner, E. (2003). Biodegradation of soil-derived dissolved organic matter as related to its properties. *Geoderma*, 113(3–4), 273–291. https://doi.org/10.1016/S0016-7061(02)00365-8
- Keiluweit, M., Nico, P. S., Kleber, M., & Fendorf, S. (2016). Are oxygen limitations under recognized regulators of organic carbon turnover in upland soils? *Biogeochemistry*, 127(2), 157–171. https://doi.org/10.1007/s10533-015-0180-6
- Kirk, T. K., & Farrell, R. L. (1987). Enzymatic "combustion": The microbial degradation of lignin. Annual Review of Microbiology, 41, 465–505. Könönen, M., Jauhiainen, J., Laiho, R., Kusin, K., & Vasander, H. (2015). Physical and chemical properties of tropical peat under stabilised land uses. Mires and Peat, 16, 1–13.
- Könönen, M., Jauhiainen, J., Laiho, R., Spetz, P., Kusin, K., Limin, S., & Vasander, H. (2016). Land use increases the recalcitrance of tropical peat. Wetlands Ecology and Management, 24, 717–731. https://doi.org/10.1007/s11273-016-9498-7
- Könönen, M., Jauhiainen, J., Straková, P., Heinonsalo, J., Laiho, R., Kusin, K., et al. (2018). Deforested and drained tropical peatland sites show poorer peat substrate quality and lower microbial biomass and activity than unmanaged swamp forest. *Soil Biology and Biochemistry*, 123, 229–241. https://doi.org/10.1016/j.soilbio.2018.04.028
- Kuznetsova, A., Brockhoff, P. B., & Christensen, R. H. B. (2017). ImerTest package: Tests in linear mixed effects models. *Journal of Statistical Software*, 82, 1–26. https://doi.org/10.18637/jss.v082.i13
- Lähteenoja, O., & Page, S. (2011). High diversity of tropical peatland ecosystem types in the Pastaza-Marañón basin, Peruvian Amazonia. Journal of Geophysical Research, 116, G02025. https://doi.org/10.1029/2010JG001508
- Lähteenoja, O., Ruokolainen, K., Schulman, L., & Oinonen, M. (2009). Amazonian peatlands: An ignored C sink and potential source. *Global Change Biology*, 15(9), 2311–2320.
- Laine, A. M., Mäkiranta, P., Laiho, R., Mehtätalo, L., Penttilä, T., Korrensalo, A., et al. (2019). Warming impacts on boreal fen CO<sub>2</sub> exchange under wet and dry conditions. *Global Change Biology*, 25(6), 1995–2008. https://doi.org/10.1111/gcb.14617
- LaRowe, D. E., & Van Cappellen, P. (2011). Degradation of natural organic matter: A thermodynamic analysis. *Geochimica et Cosmochimica Acta*, 75(8), 2030–2042. https://doi.org/10.1016/j.gca.2011.01.020
- Lehmann, J., & Kleber, M. (2015). The contentious nature of soil organic matter. *Nature*, 528, 60–68. https://doi.org/10.1038/nature16069
- Leifeld, J., Steffens, M., & Galego-Sala, A. (2012). Sensitivity of peatland carbon loss to organic matter quality. *Geophysical Research Letters*, 39, 1–6. https://doi.org/10.1029/2012GL051856
- Lindgren, A., Hugelius, G., & Kuhry, P. (2018). Extensive loss of past permafrost carbon but a net accumulation into present-day soils. *Nature*, 560, 219–222. https://doi.org/10.1038/s41586-018-0371-0
- Loisel, J., van Bellen, S., Pelletier, L., Talbot, J., Hugelius, G., Karran, D., et al. (2017). Insights and issues with estimating northern peatland carbon stocks and fluxes since the Last Glacial Maximum. *Earth-Science Reviews*, *165*, 59–80.
- Loisel, J., Yu, Z., Beilman, D. W., Camill, P., Alm, J., Amesbury, M. J., et al. (2014). A database and synthesis of northern peatland soil properties and Holocene carbon and nitrogen accumulation. *The Holocene*, 24(9), 1028–1042.
- Malhotra, A., Brice, D. J., Childs, J., Graham, J. D., Hobbie, E. A., Vander Stel, H., & Iversen, C. M. (2020). Peatland warming strongly increases fine-root growth. Proceedings of the National Academy of Sciences, 117(30), 17627–17634.
- Mathijssen, P. J. H., Gałka, M., Borken, W., & Knorr, K. H. (2019). Plant communities control long term carbon accumulation and biogeochemical gradients in a Patagonian bog. *The Science of the Total Environment*, 684, 670–681. https://doi.org/10.1016/j.scitotenv.2019.05.310
- McPartland, M. Y., Montgomery, R. A., Hanson, P. J., Phillips, J. R., Kolka, R., & Palik, B. (2020). Vascular plant species response to warming and elevated carbon dioxide in a boreal peatland. *Environmental Research Letters*, 15. https://doi.org/10.1088/1748-9326/abc4fb
- Meehl, G. A., Stocker, T. F., Collins, W. D., Friedlingstein, P., Gaye, T., Gregory, J. M., et al. (2007). Climate change 2007: The physical science basis. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 747–846). IPCC.

- Moore, T. R., Bubier, J. L., & Bledzki, L. A. (2007). Litter decomposition in temperate peatlands: The effect of substrate and site. *Ecosystems*, 10, 949–963.
- Nichols, J. E., & Peteet, D. M. (2019). Rapid expansion of northern peatlands and doubled estimate of carbon storage. *Nature Geoscience*, 12(11), 917–921. https://doi.org/10.1038/s41561-019-0454-z
- Norby, R. J., Childs, J., Hanson, P. J., & Warren, J. M. (2019). Rapid loss of an ecosystem engineer: Sphagnum decline in an experimentally warmed bog. Ecology and Evolution, 00, 1–15. https://doi.org/10.1002/ece3.5722
- Normand, A. E. (2017). Global peatland soil organic carbon chemical composition and greenhouse gas production. (Doctoral Dissertation). University of Florida.
- Page, S. E., Rieley, J. O., & Banks, C. J. (2011). Global and regional importance of the tropical peatland carbon pool. *Global Change Biology*, 798–818. https://doi.org/10.1111/j.1365-2486.2010.02279.x
- Painter, T. J. (1991). Lindow Man, Tollund Man, and other peat-bog bodies: The preservation and antimicrobial action of shagnan, a reactive glycuronoglycan with tanning and sequestering properties. Carbohydrate Polymers, 15(2), 123–142. https://doi.org/10.1016/0144-8617(91)90028-B
- Pebesma, E. J., & Bivand, R. S. (2005). Classes and methods for spatial data in R. R News, 5(2). https://cran.r-project.org/doc/Rnews/
- R Core Team. (2018). R: A language and environment for statistical computing. R Foundation for Statistical Computing. Retrieved from https:// www.R-project.org/
- R Core Team. (2020). R: A language and environment for statistical computing. R Foundation for Statistical Computing. Retrieved from https:// www.R-project.org/
- Roulet, N. T., Ash, R., Quinton, W., & Moore, T. (1993). Methane flux from drained northern peatlands: Effect of a persistent water table lowering on flux. *Global Biogeochemical Cycles*, 7(4), 749–769.
- Rydin, H., & Jeglum, J. (2013). The biology of peatlands. Oxford University Press.
- Scanlon, D., & Moore, T. (2000). Carbon dioxide production from peatland soil profiles: The influence of temperature, oxic/anoxic conditions and substrate. Soil Science, 165(2), 153–160.
- Scheffer, R. A., Van Logtestijn, R. S. P., & Verhoeven, J. T. A. (2001). Decomposition of *Carex* and *Sphagnum* litter in two mesotrophic fens differing in dominant plant species. *Oikos*, 92(1), 44–54. https://doi.org/10.1034/j.1600-0706.2001.920106.x
- Stoffel, M. A., Nakagawa, S., & Schielzeth, H. (2020). Part R2: Partitioning R2 in generalized linear mixed models. *bioRxiv*. https://doi.org/10.1101/2020.07.26.221168
- Tfaily, M. M., Cooper, W. T., Kostka, J. E., Chanton, P. R., Schadt, C. W., Hanson, P. J., et al. (2014). Organic matter transformation in the peat column at Marcell Experimental Forest: Humification and vertical stratification. *Journal of Geophysical Research: Biogeosciences*, 119(4), 661–675. https://doi.org/10.1002/2013JG002492
- Thomas, P. A., & Pearce, D. M. E. (2004). Role of cation exchange in preventing the decay of anoxic deep bog peat. Soil Biology & Biochemistry, 36, 23–32. https://doi.org/10.1016/j.soilbio.2003.07.003
- Tingley, M. P., & Huybers, P. (2013). Recent temperature extremes at high northern latitudes unprecedented in the past 600 years. *Nature*, 496, 201–205. https://doi.org/10.1038/nature11969
- Treat, C. C., Kleinen, T., Broothaerts, N., Dalton, A. S., Dommain, R., Douglas, T. A., et al. (2019). Widespread global peatland establishment and persistence over the last 130,000 y. Proceedings of the National Academy of Sciences, 116(11), 4822–4827.
- Treat, C. C., Wolheim, W. M., Varner, R. K., Grandy, A. S., Talbot, J., & Frolking, S. (2014). Temperature and peat type control CO<sub>2</sub> and CH<sub>4</sub> production in Alaskan permafrost peats. *Global Change Biology*, 20(8), 2674–2686. https://doi.org/10.1111/gcb.12572
- Turetsky, M. R., Benscoter, B., Page, S., Rein, G., Van Der Werf, G. R., & Watts, A. (2015). Global vulnerability of peatlands to fire and carbon loss. *Nature Geoscience*, 8(1), 11–14.
- Urbanová, Z., Straková, P., & Kaštovská, E. (2018). Response of peat biogeochemistry and soil organic matter quality to rewetting in bogs and spruce swamp forests. *European Journal of Soil Biology*, 85, 12–22. https://doi.org/10.1016/j.ejsobi.2017.12.004
- Urbina, J. C., & Benavides, J. C. (2015). Simulated small scale disturbances increase decomposition rates and facilitates invasive species encroachment in a high elevation tropical Andean peatland. *Biotropica*, 47(2), 143–151. https://doi.org/10.1111/btp.12191
- Van Breemen, N. (1995). Nutrient cycling strategies. Plant and Soil, 168(1), 321-326.
- Verhoeven, J. T. A., & Toth, E. (1995). Decomposition of Carex and Sphagnum litter in fens: Effect of litter quality and inhibition by living tissue homogenates. Soil Biology and Biochemistry, 27(3), 271–275. https://doi.org/10.1016/0038-0717(94)00183-2
- Vitt, D. H., Wieder, R. K., Scott, K. D., & Faller, S. (2009). Decomposition and peat accumulation in rich fens of boreal Alberta, Canada. *Ecosystems*, 12(3), 360–373.
- von Post, L. (1922). Sveriges Geologiska Undersöknings torvinventering och några av dess hittills vunna resultat. Jonkoping, 1922. Tryckt Hos H. Halls Boktryckeri-Aktiebolag. Retrieved from https://pub.epsilon.slu.se/8627/7/vonPost\_%5B1922%5D\_120313.pdf
- Walker, A. P., Carter, K. R., Gu, L., Hanson, P. J., Malhotra, A., Norby, R. J., et al. (2017). Biophysical drivers of seasonal variability in Sphagnum gross primary production in a northern temperate bog. Journal of Geophysical Research: Biogeosciences, 122, 1078–1097. https://doi. org/10.1002/2016JG003711
- Wang, H., Richardson, C. J., & Ho, M. (2015). Dual controls on carbon loss during drought in peatlands. *Nature Climate Change*, 5(6), 584–587. https://doi.org/10.1038/nclimate2643
- Wang, H., Tian, J., Chen, H., Ho, M., Vilgalys, R., Bu, Z. J., & Richardson, C. J. (2021). Vegetation and microbes interact to preserve carbon in many wooded peatlands. *Communications Earth & Environment*, 2(1), 1–8.
- Wilson, R. M., Hopple, A. M., Tfaily, M. M., Sebestyen, S. D., Schadt, C. W., Medvedeff, C., et al. (2016). Stability of peatland carbon to rising temperatures. *Nature Communications*, 7, 1–10. https://doi.org/10.1038/ncomms13723
- Worrall, F., Moody, C. S., Clay, G. D., Burt, T. P., Kettridge, N., & Rose, R. (2018). Thermodynamic control of the carbon budget of a peatland. Journal of Geophysical Research: Biogeosciences, 123(6), 1863–1878. https://doi.org/10.1029/2017JG003996
- Worrall, F., Moody, C. S., Clay, G. D., Burt, T. P., & Rose, R. (2017). The flux of organic matter through a peatland ecosystem: The role of cellulose, lignin, and their control of the ecosystem oxidation state. *Journal of Geophysical Research: Biogeosciences*, 122(7), 1655–1671. https://doi.org/10.1002/2016JG003697
- Yu, Z., Loisel, J., Brosseau, D. P., Beilman, D. W., & Hunt, S. J. (2010). Global peatland dynamics since the Last Glacial Maximum. Geophysical Research Letters, 37, L13402. https://doi.org/10.1029/2010GL043584
- Zeh, L., Igel, M. T., Schellekens, J., Limpens, J., Bragazza, L., & Kalbitz, K. (2020). Vascular plants affect properties and decomposition of moss-dominated peat, particularly at elevated temperatures. *Biogeosciences*, 17, 4797–4813. https://doi.org/10.5194/bg-17-4797-2020
- Zimmermann, M., Leifeld, J., & Fuhrer, J. (2007). Quantifying soil organic carbon fractions by infrared-spectroscopy. Soil Biology & Biochemistry, 39, 224–231. https://doi.org/10.1016/j.soilbio.2006.07.010